Author’s response to reviews

Title: A novel technique for repositioning a nasobiliary catheter from the mouth to nostril in Endoscopic Retrograde Cholangiopancreatography

Authors:

Seong Ji Choi (coolandy03@hanmail.net)
Jae Min Lee (jmlee1202@gmail.com)
Hyuk Soon Choi (mdkorea@gmail.com)
Eun Sun Kim (silverkes@naver.com)
Bora Keum (borakeum@hanmail.net)
Yeun Seok Seo (gandorie@gmail.com)
Yoon Tae Jeen (ytjeen@korea.ac.kr)
Hong Sik Lee (hslee60@korea.ac.kr)
Hoon Jai Chun (drchunhj@gmail.com)
Soon Ho Um (umsh@korea.ac.kr)
Chang Duck Kim (kumcge@korea.ac.kr)
Chi Hyuk Oh (coolsoulmate@naver.com)

Version: 1 Date: 03 Jul 2019

Author’s response to reviews:

Dear Editor,

Thank you so much for your kind comments that improve the quality of our paper.

1. As you had advised, we resubmitted our paper as a Research Article.

2. We made sure that we added the correct headins in the Declarations section as you had advised.
We are looking forward to hear a good response from you and reviewers.

Thank you for your time.

Sincerely,
Jae Min Lee